Author’s response to reviews

Title: Dietary intake in undernourished adults living in Guinea-Bissau; a cross-sectional study

Authors:

Cecilie Patsche (ceciliepatsche@live.dk;patsche@ph.au.dk)
Frauke Rudolf (frauke.rudolf@clin.au.dk)
Antonio Mendes (tboni17@gmail.com)
Idalina da Cunha (clinpoctoni@gmail.com)
Victor Gomes (victorfranciscogomes@yahoo.co.uk)
Christian Wejse (wejse@dadlnet.dk)
Charlotte Jeppesen (charl.jeppesen@gmail.com)

Version: 2 Date: 31 Jan 2019

Author’s response to reviews:

Response to Editor Comments:

R1.1. In your Ethics approval and consent to participate subsection of the Declarations, please clarify whether a parent and/or legal guardian provided written informed or thumbprint on behalf of the study participants who were under the age of eighteen and unable to consent for themselves.

C1.1. A sentence has been added on ll. 293-295 stating the process of informed consent from adolescents: "Due to the non-invasive nature of the study, adolescents aged 15-17 years were permitted to give informed consent without parental/legal guardian consent (approved by the National Committee of Ethics in Health in Guinea-Bissau).". The sentence is not marked with track changes because of the second editor comment (see below).

R1.2. At this stage, please upload your manuscript as a single, final, clean version that does not contain any tracked changes, comments, highlights, strikethroughs or text in different colours.
All relevant tables/figures/additional files should also be clean versions. Figures (and additional files) should remain uploaded as separate files.

C1.2. Is uploaded with this submission.